Author’s response to reviews

Title: Menstrual Health and Hygiene among Indigenous Australian girls and women: Barriers and opportunities

Authors:

Emily Krusz (e.krusz@uq.edu.au)
Nina Hall (n.hall2@uq.edu.au)
Dani Barrington (d.j.barrington@leeds.ac.uk)
Sandra Creamer (s.creamer@uq.edu.au)
Wendy Anders (seniorprojects@natsiwa.org.au)
Minnie King (womenoncountry@bigpond.com)
Helen Martin (womenoncountry1@bigpond.com)
Julie Hennegan (jhenneg1@jhu.edu)

Version: 1 Date: 17 May 2019

Author’s response to reviews:

Dear Editor (17 May),

We have further revised our article. Please find it attached. The additional author (Helen Martin) has been added to the manuscript, but we did not receive the form to add her as an author using the form. Please send it again.

Dear Editor (11 May),

Thank you for sharing the views of the peer reviewers for our article.

In the table attached as 'supplementary materials', we have responded to each of the comments, and endeavoured to incorporate the reviewers’ recommendations into our revised manuscript. Overall, our main and significant change is that we have restructured the article as a ‘Debate’ article. This has addressed the majority of the comments.
Our changes are visible in the ‘track changes’ version, and have been accepted into our ‘clean copy’ version of the manuscript.

We look forward to moving towards manuscript acceptance and publication.

Yours sincerely,

The authors